Author's response to reviews

Title: Factors affecting medical students in formulating their specialty preferences in Jordan

Authors:

Yousef Khader (yousef.k@exite.com)
Dema Al-Zoubi (yousef.k@exite.com)
Zouhair Amarin (zoamarin@hotmail.com)
Ahmad Kafajei (kafajei@just.edu.jo)
Mohammad Khasawneh (deema321@yahoo.com)
Samar Burgan (samarburgan@hotmail.com)
Khalid El Salem (khalidelsalem@hotmail.com)
Mousa Omari (mousa_om@just.edu.jo)

Version: 3 Date: 1 May 2008

Author's response to reviews:

The authors have replaced the word choice with preference all through the text as suggested. The figures will be uploaded separately.